Author's response to reviews

Title: Comparative health systems research in a context of HIV/AIDS: Lessons from a multi-country study in South Africa, Tanzania and Zambia

Authors:

Suraya Dawad (dawads1@ukzn.ac.za)
Nina Veenstra (veenstran@ukzn.ac.za)

Version: 4 Date: 29 October 2007

Author's response to reviews:

A title page has been added including:
1. Title of the article
2. List of authors as per the submission
3. Institutional affiliation (the same for both authors)
4. Email for the corresponding author